Author's response to reviews

Title: Variants in KCNQ1 increase type II diabetes susceptibility in South Asians
A study of 3,310 subjects from India and the US

Authors:

Latonya F Been (Latonya-been@ouhsc.edu)
Sarju Ralhan (sarjuralhan@yahoo.com)
Gurpreet S Wander (drgswander@yahoo.com)
Narinder K Mehra (narin98@hotmail.com)
JaiRup Singh (jairup@gmail.com)
John J Mulvihill (John-Mulvihill@ouhsc.edu)
Christopher E Aston (Chris-Aston@ouhsc.edu)
Dharambir K. Sanghera (Dharambir-Sanghera@ouhsc.edu)

Version: 7 Date: 21 January 2011

Author's response to reviews: see over
January 21, 2011

Journal: BMC Medical Genetics
The BMC Series Editorial Production Team
Reference: MS ID : 1763327830445086

Title: Variants in KCNQ1 increase type II diabetes susceptibility in South Asians: A study of 3,310 subjects from India and the US

Dear Editor,

Thank you for sending us the final reformatted version. We have detected some typographical errors and have re-uploaded the final version. We confirm that we do not wish to make any further changes after this submission.

Sincerely,

Dharambir K. Sanghera, PhD, FSB
Associate Professor of Human Genetics
Department of Pediatrics, Section of Genetics
University of Oklahoma Health Sciences Center
940 Stanton L. Young Blvd., Rm. D317 BMSB
Oklahoma City, OK 73104
Phone: 405-271-6026
Fax : 405-271-6027
Email: dharambir-sanghera@ouhsc.edu